Author's response to reviews

Title: Severe imported falciparum malaria among adults requiring Intensive Care: a cohort study.

Authors:

Michael Marks (michael.marks@nhs.net)
Margaret Armstrong (margaret.armstrong@uclh.nhs.uk)
Muhiddin M Suvari (muhid.suvari@uclh.nhs.uk)
Steve Batson (steve.batson@uclh.nhs.uk)
Christopher JM Whitty (Christopher.Whitty@lshtm.ac.uk)
Peter L Chiodini (Peter.Chiodini@lshtm.ac.uk)
Geoff Bellinghan (geoff.bellinghan@uclh.nhs.uk)
Justin F Doherty (tom.doherty@uclh.nhs.uk)

Version: 3 Date: 25 October 2012

Author's response to reviews:

Following comments from the editor the following changes have been made to the manuscript (25/10/12).

1) As requested Table 4 has been removed and the text of the manuscript adjusted accordingly.

2) I have reworded the section on ethics to make clear that ethics approval for the study was granted by the HTD Audit and Research committee - but that they stated that individual patient consent was not required for this study as this was a retrospective case note review using routinely collected patient clinical data.

No other changes have been made to the manuscript.

Best Wishes,

Michael Marks